Author’s response to reviews

Title: A metasynthesis of qualitative studies regarding opinions and perceptions about barriers and determinants of health services’ accessibility in economic migrants

Authors:

Andrés A Agudelo-Suárez (oleduga@gmail.com)
Diana Gil-González (diana.gil@ua.es)
Elena Ronda-Pérez (elena.ronda@ua.es)
Carmen Vives-Cases (carmen.vives@ua.es)
John G Love (j.g.love@rgu.ac.uk)
Peter Wimpenny (p.wimpenny@rgu.ac.uk)

Version: 2 Date: 14 June 2012

Author’s response to reviews: see over
Dear Editors

Please find attached the paper entitled: "A metasynthesis of qualitative studies regarding opinions and perceptions about barriers and determinants of health services’ accessibility in economic migrants", authors, Andrés A. Agudelo-Suárez, Diana Gil-González, Elena Ronda-Pérez, Carmen Vives-Cases, John G. Love & Peter Wimpenny, so that it may be considered for evaluation and possible publication in BMC Health Services Research.

For recommendation to the editorial office of BMC Public Health, it has been considered BMC Health Services Research could be more appropriated to publish the results of this systematic review and qualitative metasynthesis.

Also I am detailing how it has responded to the editorial comments.

I hope all the information may be clear and the comments of the editorial team are responded.

Kind regards,

Andrés A. Agudelo-Suárez
Corresponding author
Responses to the editorial office

Manuscript title: A metasynthesis of qualitative studies regarding opinions and perceptions about barriers and determinants of health services’ accessibility in economic migrants

ID: 1372958277456908

Date: June/2011

Please adhere to PRISMA guidelines (http://www.prisma-statement.org/) for reporting systematic reviews.

We have reviewed all the manuscript and we have adhered to the PRISMA guidelines for reporting systematic reviews. (We specified this recommendation in the methods section):

This paper has been elaborated considering the PRISMA guidelines for reporting systematic reviews [26].

The reference [26] is the following: Urrutia G, Bonfill X: [PRISMA declaration: a proposal to improve the publication of systematic reviews and meta-analyses]. Med Clin (Barc), 135(11):507-511.

We also are uploading the PRISMA check list in order to assess the quality of the items of the systematic review (supplementary material).